Author's response to reviews

Title: Screening for hemoglobin Bart's disease among fetuses at risk at mid-pregnancy using the fetal cardiac diameter to biparietal diameter ratio

Authors:

Nopamas Thathan (mamoo_a@hotmail.com)
Kuntharee Traisrisilp (doi_pui@hotmail.com)
Suchaya Luewan (suleuwan@med.cmu.ac.th)
Kasemsri Srisupundit (ksrisupu@mail.med.cmu.ac.th)
Fuanglada Tongprasert (fpunyat@mail.med.cmu.ac.th)
Theera Tongsong (ttongson@mail.med.cmu.ac.th)

Version: 3 Date: 26 March 2014

Author's response to reviews:

Response 2

MS: 6831507106418780

Fetal cardiac diameter to biparietal diameter ratio as a predictor of hemoglobin Bart's disease among fetuses at risk at mid-pregnancy

Nopamas Thathan, Kuntharee Traisrisilp, Suchaya Luewan, Kasemsri Srisupundit, Fuanglada Tongprasert and Theera Tongsong

Dear Prof Tongsong,

Your manuscript has now been reviewed by the editor and his comments are below:

Editor's Comments:
-----------------------------------------------------------------------------------
This report describes a screening test that can be utilized by general practitioners, lacking the expertise of perinatal ultrasonographers. This is now clearly indicated in the text. However, this should be clearly indicated in the Title. The title could be something like: ?Screening for hemoglobin Bart's disease among fetuses at risk at mid-pregnancy using the fetal cardiac diameter to biparietal diameter ratio

RESPONSE: The title has been changed as suggested.

-----------------------------------------------------------------------------------

Additional editorial request:

Competing Interests
Please be advised that manuscripts must include a Competing interests section. This should be placed after the Conclusions/Abbreviations. If there are none to declare, please include the statement The authors declare that they have no competing interests. Please consider the following questions and include an appropriate declaration of competing interests in your manuscript:

RESPONSE: The “Competing interests” has been added at the end of the article.

Financial competing interests

? In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.

? Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.

? Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.

? Do you have any other financial competing interests? If so, please specify.

Non-financial competing interests

? Are there any non-financial competing interests (political, personal, religious, ideological, academic, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.

For more information please visit the instructions for authors on the journal’s website.

Authors’ Contributions

For manuscripts with more than one author, all BMC Series journals require an Authors’ Contributions section to be placed after the Competing Interests section.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. To qualify as an author one should 1) have made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; 2) have been involved in drafting the manuscript or revising it critically for important intellectual content; and 3) have given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content. Acquisition of funding, collection of data, or general supervision of the research group, alone, does not justify authorship.

We suggest the following format (please use initials to refer to each author’s contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the
immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.

Contributors who do not meet the criteria for authorship should be listed in an acknowledgements section. Examples of those who might be acknowledged include a person who provided purely technical help, writing assistance, or a department chair who provided only general support.

RESPONSE: The “Authors' contributions” has been added at the end of the article.

Copyedit

After reading through your manuscript, we feel that the quality of written English needs to be improved before the manuscript can be considered further.

We advise you to seek the assistance of a fluent English speaking colleague, or to have a professional editing service correct your language. Please ensure that particular attention is paid to the abstract.

For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (http://www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq.

RESPONSE: The manuscript has been now copyedited by the native-English consultant of our faculty. However, if the editor feels that it need edited by BioMed Central (Edanz) we are happy to comply.

Name of ethics committee:

Please update your ethics statement to include the name of the ethics committee that approved your study.

RESPONSE: The ethics statement has been updated, as indicated in the “Materials and Methods” section.